Author's response to reviews

Title: Standardization of Surface Electromyography Utilized to Evaluate Patients with Dysphagia

Authors:

Michael Vaiman (vaimed@yahoo.com)

Version: 2 Date: 16 May 2007

Author's response to reviews:

Following the formatting changes request, the following changes have been made:
1. Title - presented as a sentence, no capitals
2. Qualifications - MD PhD - removed
3. Headings - sentence case
4. Keyword section - removed
5. Introduction renamed as Background
6. No underlines
7. Figure legends were moved to a position after the references
8. Figure 2 - eyes obscured